Pbar Department Meeting
Acting Department Head: Paul Derwent (Elvin on Lab Business)

Shutdown 
· Injection Area

· Biggest hold up – kicker – 

· Graphite coating on ceramic on inside of kicker

· need a special kit.to apply the coating 

· By next week should have all pieces, matter of welding up

· Alignment

· D/A

· Quad stands

· All in, all welding done, all final aligned, shunts all connected

· Most still need things done.

· Most BPMs lose.

· Polarities should be checked

· LCW connection problems, water leaks, etc...

· Some stands need electronics connected, etc…

· Target Station

· PMAG skid stripped down, week from tomorrow back together

· Dump system

· LENS and pulsed magnet

· ISEP1V

· Excessive ground current

· Instrumentation

· Flying wires buttoned up

· Calibrating Debuncher loss monitors (finished sector 30 and 20), today move to sector 50.

· Long term loss monitors for sector 30

· Poor ground

· Raised thresholds very high

· BNC to Limo adapters replaced, and noise gone.

· 60 and 10 sector BLMs were not working in Accumulator

· RF

· Replaced all PA tubes for rotators

· DRF1-6 and 1-7 still need tubes

· 3 large aperture Toroids – where to put them (Expect them Nov. 1st)?

· Q704

· Q724

· D/A upstream of SEM806

· Tours A10-A30

· Alignment of AP1?

· HV100 tomorrow

· VT108 not scheduled yet

· Beam stop work US AP2 work…
· Scissors table to lower quads has hydraulic leak, hard to repair, have to use an alternate method

· IQ1 partially out using other means, having trouble pulling all the way out.

· May have to get ceiling certified to lower from hooks in the ceiling

· Soonest quads can be out is late tomorrow, alignment crew needed no earlier until Monday.


· DEX Bump

· Keith writing a document.

· Next week, we revisit the startup discussion…

· 2 weeks to starting routine luminosity.

· End of shutdown, Bob Mau defines as end of work Friday, November 20th.

· We close up on the 15th?

